

what do we know about the effects of macroprudential policy?				
Gabriele Galati and Richhild Moessner *				
* Views expressed are those of the authors and do not necessarily reflect official positions of De Nederlandsche Bank.				

What do we know about the effects of macroprudential policy?*

Gabriele Galati^a and Richhild Moessner^{a, b}

^a De Nederlandsche Bank, the Netherlands ^b Cass Business School, United Kingdom

15 September 2014

Abstract

The literature on the effectiveness of macroprudential policy tools is still in its infancy and has so far provided only limited guidance for policy decisions. In recent years, however, increasing efforts have been made to fill this gap. Progress has been made in embedding macroprudential policy in theoretical models. There is increasing empirical work on the effect of some macroprudential tools on a range of target variables, such as quantities and prices of credit, asset prices, and on the amplitude of the financial cycle and financial stability. In this paper we review recent progress in theoretical and empirical research on the effectiveness of macroprudential instruments.

Keywords: Macroprudential policy, financial regulation.

JEL classifications: E58, G28.

_

^{*} The views expressed are those of the authors and should not be taken to reflect those of De Nederlandsche Bank. We would like to thank Bill Allen, Jan Kakes, Anton Korinek, Pierre Lafourcade, Enrique Mendoza, Lev Ratnovski, Jan Willem van den End, Peter van Els, Birger Vikoren, Peter Wierts, and participants at the Norges Bank Financial Markets Day and the SNS Conference on Financial Stability for helpful c*omments and discussion

"The application of these so-called macroprudential policies is still very much at the developmental stage. When the Bank gained operational responsibility for monetary policy in 1997, there was a long history of practical experience, together with a vast theoretical and empirical literature for us to draw on. That didn't make setting monetary policy easy, but it certainly helped. By comparison, we are still in the Stone Age in respect of deploying macroprudential policies. There is lots of scope for academia to help us out here, on both the theoretical and empirical fronts." (Bean, 2012)

1. Introduction

The global financial crisis has highlighted the need to go beyond a purely micro-based approach to financial regulation and supervision. With the benefit of hindsight, there has been a fundamental lack of understanding of system-wide risk.

There is a growing consensus among policymakers that a macroprudential approach to regulation and supervision should be adopted. Policymakers around the globe are currently working on implementing macroprudential policy tools and frameworks. These efforts involve coordination among central banks, governments and regulatory authorities (which may also be the central bank). At the same time, analysis is still needed about the appropriate macroprudential tools, their transmission mechanism and their effects.

Theoretical models of macroprudential policy are still in their infancy and therefore provide only limited guidance on how to think about its impact. However, major efforts have been undertaken over the past years to pursue this line of research. Empirical evidence on the effects of macroprudential instruments is still scarce, given that they have become standard policy tools only in recent years. In advanced countries, the experience with macroprudential policy is fairly recent, although some measures to support the domestic financial system and to influence the supply of credit taken in the 1930s and 1950s have been viewed as macroprudential tools (Haldane, 2011). By contrast, central banks in emerging market countries have been regular practitioners of macroprudential policy, without calling it by this name (McCauley, 2009).

This paper aims to tie together theoretical and empirical contributions to the literature on macroprudential policy. The aim is to provide guidance on how to think about the effectiveness of macroprudential tools and how they are transmitted to financial intermediation and macroeconomic activity. In our analysis we distinguish two forms of effectiveness – in achieving stated intermediary goals of macroprudential policy (e.g. credit growth), and in achieving its ultimate goal of supporting financial stability and thereby macroeconomic stability.

The remainder of the paper is organized as follows. Section 2 discusses the perimeter and taxonomies of macroprudential instruments. Section 3 provides an overview of their usage, including an overview of the historical experience. Section 4 review alternative approaches to conceptualize the effectiveness and transmission mechanism of macroprudential tools. It goes on to discuss the evidence on the effectiveness of macroprudential tools. Section 5 discusses issues affecting the effectiveness of macroprudential instruments, and section 6 concludes.

2. What are macroprudential instruments?

Macroprudential policy can best be understood by contrasting it with the traditional microprudential perspective on regulation. The micro and macroprudential perspectives differ in terms of their objectives and understanding on the nature of risk (Borio, 2003). The former aims at enhancing the safety and soundness of individual financial institutions. The latter instead focuses on the stability of the financial system as a whole, with a view to limiting macroeconomic costs from financial distress (Crockett, 2000). Another key difference is that risk is taken as exogenous under the microprudential perspective, while the macroprudential perspective emphasizes the endogenous nature of systemic risk. Macroprudential policy therefore focuses on the procyclical behavior of the financial system, and the interconnectedness of individual financial institutions and markets, as well as their common exposure to economic risk factors.

In contrast to the literature on monetary policy with its clear-cut consensus on the role of different instruments, the literature on macroprudential policy has looked at a wide range of possible tools without a primary instrument emerging. It has only recently been converging towards a common understanding of its perimeter and a common taxonomy of instruments.

In terms of the perimeter of macroprudential policy, macroprudential instruments are commonly viewed as distinct from other macroeconomic tools that can support financial stability, such as monetary and fiscal policy (Blanchard et al, 2010; Borio, 2009; Caruana, 2010). A combination of these tools is seen as necessary to address systemic risk in all its complexity (Caruana, 2010).

We follow IMF (2011a) and define the macroprudential toolbox as comprising two types of instruments. One is specifically tailored to mitigating systemic risk, and includes countercyclical capital buffers or systemic capital charges. The other is instruments not originally developed with systemic risk in mind but that can be modified to become part of the macroprudential toolkit. Timevarying loan-to-value (LTV), loan-to-income (LTI) or debt-to-income (DTI) ratios belong to this category. IMF (2011a) argues that for these to be macroprudential tools, two conditions need to hold. First, they should target explicitly and specifically systemic risk. Second, the chosen institutional framework is underpinned by the necessary governance arrangements to ensure there is no slippage in their use.

While precise in theory, the delimitation of the perimeter of macroprudential policy is not clear-cut in practice. Recent surveys conducted among central banks reveal that there are grey areas in regard to what counts as a macroprudential instrument (CGFS, 2010). Most notably, there is no agreement on whether capital controls should be considered as part of the macroprudential arsenal. The FSB and BIS for example do not include capital controls among macroprudential policy tools, since they are applied only to non-residents, whereas some countries' policymakers consider them to be macroprudential instruments. Following IMF (2011b), capital controls could be macroprudential instruments if they are geared towards systemic risk and are underpinned by strict governance arrangements. Given the historical experience of build-ups of financial imbalances accompanied by a growing share of net foreign-currency financing (Borio and Shim, 2007), especially for emerging market economies, the macroprudential toolkit could also include measures to limit system-wide currency mismatches in an effort to stem the domestic financial consequences of capital inflows. Examples of such measures include limits on open foreign exchange positions and constraints on the type of foreign currency assets (Turner, 2009).

Another grey area is measures geared towards strengthening the resilience of the infrastructure of the financial system. It is now common to include these in the macroprudential toolkit (e.g. Group of Thirty, 2010; BIS-FSB-IMF, 2011; Schoenmaker and Wierts, 2011).

A third grey area is crisis resolution. While this policy area also focuses on systemic risk it involves more reactive rather than preventative policy and thus typically is not included in the macroprudential toolkit (see e.g. IMF, 2013; Osiński et al., 2013).

In the policy debate and the research literature, it is possible to identify two main taxonomies of macroprudential tools, which help in understanding how they affect the financial system and the macroeconomy.¹

Nature of risk and intermediate objectives

The first organizes tools in terms of the nature of systemic risk they address and what specific objectives they are geared towards. In terms of the nature of systemic risk, the main and by now common distinction is between tools geared towards addressing the time dimension and those focusing

¹ Distinctions along other dimensions have also been used, including rules (built-in stabilizers) versus discretion in calibrating the tools of macroprudential policy (e.g. Borio and Shim, 2007); tools based on quantity or on price restrictions (e.g. Perotti and Suarez, 2011); and instruments used in industrial or emerging market countries (McCauley, 2009).

Table 1 Macroprudential tools across different risk dimensions and intermediate objectives

Intermediate objectives/ functions	Risk dimensions	
	Time-series	Cross-sectional
Leverage/credit/asset price booms	Countercyclical capital buffers	
	Through-the-cycle valuation of margins or haircuts for collateral used in securitized funding markets (like repos)	
	Countercyclical change in risk weights for exposure to certain sectors	
	Time-varying LTV, Debt-To-Income (DTI) and Loan-To-Income (LTI) caps	
	Time-varying caps and limits on credit or credit growth	
	Dynamic provisioning	
	Rescaling risk-weights by incorporating recessionary conditions in the probability of default assumptions (PDs)	
Liquidity/market risk	Time-varying systemic liquidity surcharges	Capital charge on derivative payables
	Levy on non-core liabilities	Levy on non-core liabilities
	Time-varying limits in currency mismatch or exposure (e.g. real estate)	
	Time-varying limits on loan-to-deposit ratio	
	Stressed VaR to build additional capital buffer against market risk during a boom	
Interconnectedness/ market structure/financial infrastructrure		Higher capital charges for trades not cleared through CCPs
		Systemic capital surcharges
		Systemic liquidity surcharges
		Powers to break up financial firms on systemic risk concerns
		Deposit insurance risk premia sensitive to systemic risk
		Restrictions on permissible activities (e.g. ban on proprietary trading for systemically important banks)

Sources: IMF (2011a; 2013). Entries include a) instruments developed specifically to mitigate systemic risk and b) instruments not originally aimed at systemic risk but modified to be part of macroprudential toolkit (in italics). The classification in terms of intermediate objectives is taken from Schoenmaker and Wierts (2011) and Houben et al. (2012).

on the cross-sectional dimension of systemic risk.² The former address the procyclicality in the financial system. The latter focus on how risk is distributed at a point in time within the financial system and on contributions to systemic risk of individual institutions.

Within each dimension of systemic risk, it is then possible to identify intermediate objectives to which macroprudential tools are assigned and specific functions for each tool.³ In terms of specific functions or intermediate objectives to which macroprudential tools are assigned, the policy debate seems to converge to distinguishing three types of tools (Table 1).⁴

The first type of tools focuses on the time dimension of systemic risk, and comprises tools geared towards credit, leverage and asset price booms. The second is also geared towards the time dimension, and addresses liquidity or market risk. The third type of tools addresses vulnerabilities that arise from either market structure – i.e. vulnerabilities related to interconnectedness, size, position in the market – and those originating in the financial infrastructure (e.g. related to central counterparty clearing and real time gross settlement systems).⁵ These structural vulnerabilities pertain mainly to the crosssection dimension of systemic risk.

Note that tools that address the time dimension and the cyclical nature of systemic risk need not be time-varying. Examples of time-invariant instruments of this type include fixed caps on cyclical variables such as LTV limits or loan-to-deposit (LTD) ceilings. On the other hand, tools that address the cross-sectional dimension of systemic risk may not be time-invariant (see BCBS, 2011).

Macroprudential tools geared towards market failures

An alternative taxonomy focuses on the market failures that macroprudential policy addresses. As De Nicolò et al. (2012) stress, addressing the time and the cross-section dimensions of systemic risk per se does not provide a justification for regulatory intervention. Rather, macroprudential regulation should be viewed as addressing market failures that create systemic risk. These market failures include risk externalities across financial institutions, and between the financial sector and the real economy.

The literature distinguishes three main drivers of such risk externalities (De Nicolò et al., 2012): strategic complementarities, i.e. strategic interactions of financial institutions that cause the build-up of vulnerabilities during the expansionary phase of the financial cycle (herding); fire sales, i.e. a generalized sell-off of financial assets, which cause a decline in asset prices and a deterioration of the balance sheets of financial intermediaries; and interconnectedness, i.e. the risk of contagion caused by the propagation of shocks from systemic institutions or through financial networks. As noted by De Nicolò et al. (2012) externalities driven by interconnectedness are particularly strong for SIFIs.

An influential example of research along these lines is Hanson et al (2011), which characterizes the macroprudential approach to capital regulation as an effort to control the social costs associated with excessive balance-sheet shrinkage on the part of multiple financial institutions hit with a common shock. To achieve this, there are only two options: raising new capital to replace that which was lost;

² This distinction was first introduced by Borio and Crockett (2000).

³ The concept of intermediate objectives – which is commonly employed in the monetary policy literature – has been introduced in the literature on macroprudential policy by Schoenmaker and Wierts (2011).

⁴ This distinction is elaborated in IMF, 2011a,b; FSB-BIS-IMF, 2011b; CGFS, 2010; Bank of England, 2011; ESRB, 2011; Houben et al., 2012. A similar taxonomy is provided by CGFS (2010), which groups macroprudential tools depending on the vulnerability they address - leverage, liquidity or market risk, interconnectedness - and on the financial sector component they are geared to (bank or deposit-taker, non-bank investor, securities market or financial infrastructure). IMF (2011a,b) also highlight those associated with capital flows and those related to foreign currency risk.

⁵ ESRB (2011) refines this third set of tools by differentiating instruments that address risk arising from (direct and indirect) exposures, risk related to expectations of a bailout, and risk linked to "soft" infrastructure (such as accounting practices or risk management standards) and "hard" infrastructure (e.g. netting requirements in settlement systems).

or letting the ratio of capital to assets decline. The tools that Hanson et al. discuss are different mechanisms for facilitating adjustment on one of these two margins.⁶

Macroprudential instruments can then be classified according to the externalities they can address. Table 2, taken from De Nicolò et al. (2012), provides such a classification and indicates that alternative policy tools are often complementary. This suggests that there is no one-to-one mapping between externality and instrument, and that a combination of tools is likely to be more effective in tackling a market failure. It also suggests that among all macroprudential instruments, capital surcharges are effective in dealing with all of the externalities.

Table 2 Macroprudential tools across different types of market failures/externalities

Externalities	Instruments				
	Capital requirements (surcharges)	Liquidity requirements	Restrictions on activities, assets, or liabilities	Taxation	
Strategic complementarities	Х		Х		
Fire sales	X	x		X	
Interconnectedness	X		х	X	
Source: De Nicolò, Favara and Ratnovski (2012).					

This approach has been followed also in practice. A notable example is the European Systemic Risk Board (ESRB), which is tasked with macroprudential oversight of the financial system within the European Union. The ESRB maps market failures – including externalities such as those related to fire sales and interconnectedness – into intermediate objectives for macroprudential policy (ESRB, 2014a, 2014b).

3. Experiences with macroprudential instruments

Since the outbreak of the global financial crisis, macroprudential policy has come to play a key role in the policy debate but the experience with macroprudential instruments goes back as far as the 1930s. Some of the tools which are currently being used or proposed as part of the macroprudential toolset were originally used with microprudential objectives. Others were seen as monetary policy instruments to influence the supply of credit and growth (Bank of England, 2009), often focused on macroeconomic stabilization in the presence of large capital flows and a volatile exchange rate (Cordella et al., 2014).

In discussing experiences with macroprudential policy, it is useful to distinguish these historical experiences further in the past from recent experiences. Haldane (2011) identifies a first phase in the 1930s, when policymakers adopted a variety of measures to support the domestic financial system and to influence the supply of credit. According to Haldane (2011), macroprudential policy was used as early as 1938 in the United States, in a successful effort by Roosevelt to boost lending and growth, as the country was facing a double-dip recession, with banks being criticized for not lending more to the real economy. These measures took the form of a relaxation of prudential and valuation standards for US banks in the Uniform Agreement on Bank Supervisory Procedures, whose explicit goal was to support lending and activity in the real economy, "the activist goal of liberalizing bank examinations

_

⁶ These tools are grouped in six sets that comprise time varying capital requirements; higher quality capital; prompt corrective action targeted at dollars of capital, not capital ratios; contingent capital; regulation of debt maturity and asset liquidity; regulating the shadow banking system.

to make them dynamically adjustable to current economic policies" (Simonson and Hempel, 1993, as cited in Haldane, 2011).

Following this interpretation, one could view restrictions introduced on both the asset side and the liability side of US banks' balance sheets in the 1930s and in the following decades – in an effort to support the domestic banking system – as macroprudential measures. In the United State, these include interest rate ceilings, interstate banking restrictions, and the Glass-Steagall separation of commercial and investment banking. They also include selective credit controls introduced in the 1950s with the aim of influencing the housing cycle (Grebler, 1960) and used throughout the 1970s (Schreft, 1990). Elliott et al. (2013) provide a detailed review of the historical usage of and provide a qualitative assessment of the costs and benefits of countercyclical macroprudential tools used in the United States since the First World War. They suggest that many of these tools appear to have succeeded in their short-term goals, for example in limiting specific types of bank credit or liability and affecting lending terms, while it was less obvious that they improved financial stability.

Similar measures were taken in the 1960s and 1970s in other countries. Well-known examples are the direct credit ceilings and the special deposit scheme known as "the corset" introduced in the United Kingdom in an effort to counter the rapid growth in domestic banks' exposures and its destabilizing effect. Similar policies were also present in the Netherlands between the 1960s and 1980s, where different forms of instruments were employed to contain credit creation by banks (de Greef et al., 1996). Another example is Sweden, where the Swedish Riksbank introduced domestic credit controls in the 1950s, which were supported by exchange controls (Jonung, 1993). It has been argued that these policies helped prevent financial crises (Englund, 1999). Most of these measures were phased out in the 1980s and 1990s, as deregulation and globalization changed the global financial landscape, reflecting policymakers' concerns over the viability of traditional banking in the face of pressure from non-bank financial institutions (Hellwig, 1994, 1995).

Since the 1970s, policy makers in emerging markets facing large capital flows have used reserve requirements in an effort to stabilize the domestic economy by stabilizing the exchange rate and the credit cycle. As argued by Cordella et al. (2014), the distinction between this "business-cycle driven use" of macroprudential policy and the "systemic risk-driven" use is not clear-cut, since the use of macroprudential instruments to deal with the business cycle supports financial stability by preventing excessive fluctuations in capital flows and hence reduces the probability of systemic risk.

Recent experiences

In the second phase, which took place in more recent years, macroprudential tools were (re-)adopted – mostly in emerging market countries – to strengthen the resilience of the domestic financial system. Recent surveys conducted among central banks provide an overview of central banks' experience with macroprudential policy (CGFS, 2010; IMF, 2011b,c). They show that the use of macroprudential policy instruments – whether or not they were called by their name – has been most widespread in emerging market countries in Asia and Latin America. In these countries, central banks have applied macroprudential policies at least since the aftermath of the 1997 Asian financial crisis and the 1998 Russian financial crisis, in an effort to strengthen the resilience of their domestic financial system.

The most commonly used tools include measures to limit credit supply to specific sectors that are prone to excessive credit growth (e.g. caps on LTV ratios or debt/income ratios aimed at restricting mortgage lending), and limits on net open currency positions and measures to prevent the build-up of domestic imbalances arising from cross-border capital flows (for example via reserve requirements). The former were most widely used in Asia, whereas the latter could be found mostly in Latin American countries (Moreno, 2011; Montoro and Moreno, 2011).

_

⁷ Hodgman (1973) provides a survey of the use of credit control in European countries. Shu et al. (2008) draw parallels with credit controls used by the People's Bank of China in the 2000s.

⁸ IMF (2011b) surveyed central banks from 63 countries and the European Central Bank about the use of macroprudential tools in the second half of the 2000s. CGFS (2010) conducted a survey at end-2009 of 33 central banks.

Before the global financial crisis, macroprudential tools have not been much applied on a system-wide basis to banks' balance sheets in advanced countries. An important exception is dynamic provisioning in Spain, which arguably did not prevent a major bubble in the domestic housing market but made the banking system more resilient to shocks by increasing its shock-absorption capacity (Saurina, 2009a,b). However, some observers highlighted that these tools were not sufficient to prevent the large problems in the Spanish banking sector which emerged following the unwinding of the domestic real estate boom in 2007 (e.g. Mahapatra, 2012).

In recent years, measures that target the size or the composition of bank balance sheets – e.g. LTD ceilings, institution-specific capital add-ons or time-varying capital charges – have gained in importance.⁹

4. Alternative approaches to investigate the effectiveness of macroprudential policy and its transmission mechanism

It has been argued that the transmission mechanism through which macroprudential policy and monetary policy work is similar to the extent that both work through the bank lending and balance sheet channels, and both are geared towards modifying private agents' behavior (Beau et al., 2012). In fact, the literature on the monetary transmission mechanism can also offer insights on how macroprudential tools can work. In contrast to the monetary policy literature, however, the effectiveness and the transmission mechanism of macroprudential policy tools are not yet well understood. From a theoretical point of view, a main reason is that there is no agreed modeling framework of the interaction between the financial system and the macroeconomy (see Galati and Moessner, 2013). From an empirical perspective, macroprudential tools have to a large extent been introduced only in response to the recent crisis, which makes it difficult to assess their effectiveness and transmission channels empirically and guide the design of macroprudential tools going forward (see Turner, 2010).

Complicating the analysis is that macroprudential measures have typically not been taken in isolation, but in combination with other policies. Moreover, as emphasized in CGFS (2010), the transmission mechanism is likely to change over time as the result of changes in financial intermediation practices and in the structure of the financial system. In particular, there is uncertainty about how financial innovation, consolidation in the financial sector and changes in the balance between institution- and market-based credit affect systemic risk over time.

For all these reasons, the literature on the effectiveness and the transmission mechanism of macroprudential policy is still in its infancy and has so far provided only limited guidance for policy decisions. In recent years, however, much effort has been made to fill this gap. Progress has been made in embedding macroprudential policy in different types of theoretical models. There is increasing empirical work on the effect of some macroprudential tools on a range of intermediate target variables, such as quantities and prices of credit, asset prices, output growth, and on the amplitude of the financial cycle.

In this section, we review recent progress along three lines of research on the effectiveness and transmission mechanism of macroprudential instruments. The first consists of alternative theoretical frameworks for modeling macroprudential policy and its impact on financial and real variables. These include banking/finance models and macroeconomic models. The second comprises stylised presentations of how changes in individual macroprudential instruments are expected to contribute to the objectives of macroprudential policy. This analysis is typically narrative and conducted through so-called "transmission maps". The third line of research uses different strategies to empirically assess the effectiveness of macroprudential policy tools.

capital add-ons.

⁹ LTD ceilings have been introduced in China, Korea and the United States (see Federal Reserve Board, 2012; Lim et al., 2011; and van den End, 2013 for details). SIFI buffers are the most common institution-specific capital add-ons

For an overview of models of macroeconomic implications of financial frictions, see Brunnermeier et al. (2013).

a. Theoretical approaches

There exists by now a large body of research on the interaction between the macroeconomy and financial stability, which can offer insights on the possible impact of macroprudential policy. Many contributions have been reviewed elsewhere (e.g. Brunnermeier et al., 2013; Benigno, 2013; Galati and Moessner, 2013; Beck et al., 2014). Therefore, rather than providing a detailed discussion of these contributions, this section describes examples of theoretical approaches that offer insights on the impact and transmission mechanism of macroprudential tools, to give an overview of where the literature is heading. For the ease of exposition, these approaches can be grouped into three types of models: banking/finance models; three-period banking or dynamic stochastic general equilibrium (DSGE) macro models; and infinite horizon general equilibrium macro models.

Banking/finance models

Banking/finance models capture the state-dependent nature of assets and contracts (e.g. Diamond and Dybvig 1983; Diamond and Rajan, 2001). In these models, financial contracts are affected by informational asymmetries, commitment and incentives problems which can result in default. Financial instability can then result from self-fulfilling equilibria generated by exogenous shocks or from idiosyncratic or systematic shocks that propagate through the financial system through informational and balance sheet linkages.

One important advantage of these models is that they can explain the complex interaction between borrowers and lenders. They have therefore offered important insights on the impact of macroprudential policy tools that address the cross-sectional dimension of systemic risk. One important example is the paper by Perotti and Suarez (2011), which compares price-based and quantity-based regulation of systemic externalities originating in banks' short-term funding. This distinction between price- and quantity based tools plays a key role in the current policy debate. The paper shows that a combination of a Pigouvian tax on short-term funding with a net stable funding ratio or a liquidity coverage ratio might be optimal for regulators, depending on the nature of heterogeneity across banks.

At the same time, these models have two important shortcomings. First, to be tractable these models typically neglect the role of time and the business cycle, which makes it more difficult to study macroprudential tools geared towards the procyclicality of the financial system. Second, they are mostly partial equilibrium models, and recent research suggests that the effect of macroprudential tools is different in a general equilibrium setting (Al-Darwish et al., 2011; Jaffee and Walden, 2011).

Three-period general equilibrium models

Recent research has addressed these issues by developing stylized three-period general equilibrium models to study the interaction between asset prices and financial distress in the corporate and financial sector. These papers analyze the risk-taking behavior of heterogeneous agents in an economy that is vulnerable to systemic risk and in which default can occur.

One type of such models have the key feature that financial amplification during credit booms and busts involves externalities (e.g. Lorenzoni, 2008; Tsomocos, 2013; Goodhart et al., 2012; Gersbach and Rochet, 2012a, 2012b). These externalities arise because individual agents take financial decisions without taking into account the general equilibrium effect of their actions. During times of stress, agents neglect the effect of their asset sales on aggregate prices. As a result, agents take on socially excessive exposure to risk and over-borrow, eventually creating feedback loops of falling asset prices, tightening financial constraints and fire sales. Although these models are highly stylized, they can be used to study the impact of macroprudential tools geared towards preventing fire sales and credit crunches, which are found to be critical for increasing financial stability and improving the welfare of savers and borrowers (Goodhart et al., 2013). These include loan to value ratios, capital requirements for banks, liquidity coverage ratios for banks, dynamic loan loss provisioning for banks, and margin requirements on repurchase agreements used by shadow banks.

_

¹¹ For more details, see Galati and Moessner (2011).

Another type of three-period general equilibrium models captures the different roles that banks play in the economy – providing liquidity insurance for savers, improving the risk sharing opportunities for savers, increasing the amount of funding available to borrowers (Kashyap et al., 2014). These models can be used to analyze alternative causes of the global financial crisis – excessive risk-taking by under-capitalized banks which were exploiting taxpayer support; funding vulnerabilities in the financial system and bank runs – and regulatory tools that can help prevent future crises.

This research highlights how the introduction of new regulation will induce banks and savers to endogenously alter their other portfolio choices. As a result, different regulatory tools can have a very different impact on financial stability. In particular, prudential tools that moderate the risk of a run can exacerbate problems caused by limited liability. One important issue highlighted by recent research based on three-period general equilibrium models is the possibility of disturbing "interactions" between countercyclical macroprudential tools and instruments that focus on the cross-section dimension. In particular, Horváth and Wagner (2013) show that countercyclical bank regulation might increase cross-sectional risk, while policies that reduce cross-sectional risk reduce procyclicality.

Infinite horizon macroeconomic models with financial factors

Infinite horizon DSGE models augmented with financial frictions, which build on the financial accelerator mechanism of Bernanke et al. (1999), have the potential to investigate the effects of macroprudential tools that focus on the time dimension of systemic risk. Their general equilibrium nature makes them attractive for policy analysis. Frictions related to financial intermediaries and the role of bank capital in the monetary transmission mechanism have been studied with DSGE models (e.g. Goodfriend and McCallum, 2007). Moreover, they are particularly suitable for simulations, which makes them useful to study the impact of new policy instruments.

Until a few years ago, macroeconomic models of this type suffered from several important drawbacks, which limited their use in studying the effectiveness and transmission mechanism of macroprudential instruments. First, being traditionally solved by linearization, these infinite horizons models were not well suited to incorporate state-contingency in a meaningful way and hence to analyze systemic crises or the impact of changes in regulation. Second, with very few recent exceptions, these models generally assumed complete markets and implicitly assumed that defaults either do not occur or are exogenous (see e.g. Cúrdia and Woodford, 2010). Third, as argued forcefully by Geanakoplos (2011), they ignored endogenous leverage. By construction, therefore, these types of DSGE models with financial frictions describe financial crises as big negative shocks that are amplified. They cannot capture the fact that crises are rare events resulting from "credit booms gone wrong" that are followed by deep and long recessions (Boissay et al., 2013).

Since the global financial crisis, however, much effort has been undertaken to extend infinite horizon DSGE models to include a richer characterization of real-financial linkages. Compared to standard DSGE models, these models have multiple equilibria, and are used to analyze non-linearity and externalities. They feature more elaborate amplification mechanisms. These types of model can provide a framework to evaluate macroprudential (or monetary) policies which can be used during a credit boom to reduce the expected costs of a financial crisis.

One line of research that promises insights on the impact of macroprudential policy relies on infinite-horizon DSGE models with borrowers (households or firms) facing occasionally binding endogenous constraints. These models have been used to explain the mechanism of financial amplification in terms of the interaction of falling asset prices, declining net worth, tightening financial constraints and macroeconomic contraction. These models can be used to examine macroprudential policies over booms and busts and determine the optimal magnitude of specific instruments.

This line of research has been termed "Neo-Fisherian", since the financial amplification it describes captures features of Fisher's (1933) debt-deflation spiral. The borrowing constraint is occasionally binding, depending on private agents' and policymakers' choices, which determine the state of the economy (Bianchi and Mendoza, 2010; Jeanne and Korinek, 2012; Benigno et al., 2013). In crisis

_

¹² See Benigno (2013) for a detailed discussion of the modelling elements of this line of research.

times the constraint is binding, whereas in tranquil states it is not. The borrowing constraint is endogenous and depends on asset prices such as the price of land.

Another class of models studies global dynamics in models with financial frictions in a continuous time setting (He and Krishnamurthy, 2012, 2013; Brunnermeier and Sannikov, 2014; Adrian and Boyarchenko, 2012). In this research line, macroeconomic factors and the financial system are integrated in an analysis that is not confined around the steady state, which provides important insights on the mechanisms that can lead to financial crises.

In the seminal paper by Brunnermeier and Sannikov (2014), the financial sector does not internalize all the costs associated with excessive risk taking. This results in excessive leverage and maturity mismatch. While securitization allows the financial sector to offload some of the risk, it exacerbates excessive risk-taking. In this model, systemic risk and volatility dynamics are endogenous due to adverse feedback loops between financial and real factors. The model thus portrays an economy with low volatility and "reasonable" growth around the steady state but with high volatility and large output losses away from the steady state. The economy is inherently unstable because leverage and risk-taking are endogenous. An important feature of this model is the so-called "volatility paradox": as aggregate risk declines, equilibrium leverage goes up and amplification becomes more severe. The model embeds key nonlinearities: small shocks keep the economy near the stable steady state but large shocks lead to an unstable crisis regime characterized by liquidity spirals.

The classes of models with incomplete asset markets, aggregate shocks, heterogeneous agents facing occasionally binding constraints and endogenous systemic risk are challenging from a computational point of view. They rely on non-linear global solution methods to evaluate short- and long-run effects of financial frictions. These solution methods cannot be easily implemented for large models. The analysis is therefore mostly restricted to a limited set of shocks and states.

While these papers have mostly provided qualitative insights, recent research has been able to match quantitatively the nonlinearities that can be found in macroeconomic and financial data, and in particular the different dynamics across tranquil times, periods of stress and systemic crises (He and Krishnamurty, 2014).

In recent research, Benes et al. (2014a, 2014b) propose an alternative approach to include financial intermediaries' balance sheets into DSGE models to yield highly nonlinear feedback effects between bank balance sheets, borrower balance sheets and the real economy during financial crises. The MAPMOD model they introduce is a simulation model whose parameters are calibrated to match basic stylized facts of financial cycles.

This model provides a flexible tool to simulate the impact of specific macroprudential policy decisions, such as changes in countercyclical capital requirements, and a wide range of scenarios affecting the financial sector. It shares the problem faced by DSGE models with occasionally binding constraints that it cannot be estimated because of global nonlinearities and changing financial sector policies, which would require unrealistically large samples.

Main insights on macroprudential policy

While still in their infancy, the different strands that incorporate real-financial linkages into macroeconomic or banking models offer several insights to the policy debate on macroprudential instruments.

First, one main contribution of modern versions of DSGE models with financial-real linkages has consisted in explaining mechanisms through which real and financial factors interact, and how this interaction can generate systemic crises. In particular, the combination of a macroeconomic boom, a credit boom and low interest rates is conducive to a crisis when the credit boom turns supply-driven, as illustrated by Boissay et al. (2013). This research highlights the role of consumption smoothing and precautionary savings in the build-up of financial imbalances.

Second, calibrations of these models improve our understanding of the role of regulation in reducing the incidence of financial crises. Based on a quantitative analysis of their model calibrated to US data,

for example, Bianchi and Mendoza (2010) show that in the absence of regulation, financial crises are significantly more frequent and more severe.

Third, state-contingent taxes can play an important role in supporting financial stability. In Bianchi and Mendoza (2010), a regulator can replicate exactly its equilibrium allocations as a decentralized equilibrium, and thus neutralize the credit externality on which these models hinge, by imposing state-contingent taxes on debt and dividends of about 1 and -0.5 percent on average respectively. This tax is higher during periods when leverage is building up and the economy is becoming vulnerable to a financial crisis. The idea is that such as tax induces agents to value more the accumulation of precautionary savings with respect to the competitive equilibrium without taxes.

Fourth, externalities that underpin endogenous systemic risk can be addressed by Pigouvian taxes. Jeanne and Korinek (2010) for example show that a Pigouvian tax on borrowing may induce borrowers to internalize these externalities and increase welfare. More generally, these types of models allow characterizing the optimal mix of ex-ante macroprudential policy and ex-post policy (i.e. bailouts) in response to financial crises.

b. Stylized presentations

In recent years, several studies have followed an alternative, narrative approach to studying the transmission mechanism and effectiveness of macroprudential tools. An influential report by the CGFS (2012a) provides a conceptual discussion of the transmission mechanism of a range of macroprudential instruments, aiming to provide guidance on how the efficiency and effectiveness of these instruments could be judged in practice. The report presents so-called "transmission maps" for the impact on the credit cycle and on the resilience of the financial sector of stress tests, as well as tools such as capital or provisioning requirements, sectoral capital requirements, liquidity requirements and asset-side instruments. Where possible, the narrative analysis is supplemented with empirical evidence to provide some indications of the effectiveness and efficiency of different tools.

This type of analysis provides several important results. First, in terms of monitoring the build-up of financial imbalances and predicting crisis, CGFS (2012a) documents how the credit-to-GDP gap, the debt service ratio, the growth in residential property prices and their gap turn out to have been useful indicators in signaling past crises. By contrast, the predictive content of variables involving consumer credit appears to be limited.

Second, the transmission map analysis points to capital- and liquidity based macroprudential tools as playing an important role in increasing the resilience of the financial sector and smoothing the credit cycle. At the same time, these policies imply potential costs that affect their efficiency.

Third, raising capital or liquidity requirements enhances the resilience of the banking system through both direct and indirect channels. These measures also influence the shape of the credit cycle, as banks will tend to respond to a rise in generic liquidity requirements by adjusting the profile of their assets and liabilities.

Fourth, asset-side tools can play an important role in increasing the resilience of the banking system directly through decreasing both the probability of default (PD) and loss-given-default (LGD) of loans. Moreover, they have potentially significant dampening effects on the credit cycle. In particular, tighter LTV and DTI ratio caps can restrain the quantity of credit by limiting the funding available for certain borrowers, reducing housing demand and by increasing savings. These tools would tend to ease house prices, thereby reducing households' ability to obtain credit and withdrawing equity more generally. As a result, these tools will likely lead to a broad decline in the demand for credit.

Finally, this type of stylized analysis highlights the interaction between different macroprudential instruments, and between macroprudential policy and other forms of policy, such as monetary and fiscal policy.

c. Empirical approaches

Empirical analysis on macroprudential instruments is difficult because of the lack of established models of the interaction between the financial system and the macroeconomy, as well as by the

scarcity of data needed to conduct empirical tests.¹³ As discussed in Galati and Moessner (2013), until a few years ago, little analysis was available on data needs for macroprudential policy purposes. In recent years, however, much effort has been undertaken in identifying data requirements. Examples of recent proposals on data collection and requirements include Lo (2009), Sibert (2010) and Brunnermeier et al. (2014).

At the international policy level, a broad picture of data needs is emerging (see eg. Borio, 2010; Cecchetti et al., 2010; Eichner at al., 2010). Planned enhancements to the BIS international banking statistics include more granular data collected to better measure the size and volatility of cross-border borrowing by the resident non-financial private sector from non-resident banks, and how it compares with borrowing from resident banks, which would be useful for decisions on the use of macroprudential policy tools (CGFS (2012b)). There have also been efforts to build a reporting framework for systemically important banks, which provides insights on the network connections among these banks (BCBS, 2013a). The G20 Data Gaps Initiative is a broad thrust towards collecting data that are needed for policy analysis (e.g. FSB-IMF, 2013).

In terms of information on the usage of macroprudential tools, the IMF conducted a survey of country authorities on macroprudential policy actions (IMF (2011b)). Claessens and Ghosh (2012) use data on macroprudential policy actions from this survey, and supplement it with data from an internal IMF survey of country desk economists. Lim et al. (2013) combine this type of information in a so-called macroprudential index. Recent progress on data collection include Shim et al. (2013), who put together a database on policy actions related to the housing markets globally and made this publicly available. Vandenbussche et al. (2012) collected information on macroprudential policy measures related to house prices in a database for 16 countries in Central, Eastern, and South-Eastern Europe at a quarterly frequency. Federico et al. (2012a) constructed a quarterly dataset on legal reserve requirements for 52 countries, of which 15 are industrial and 37 developing countries, focussing on legal rather than actual reserve requirements, for 1970-2011. Using this dataset, they describe stylized facts regarding the use of reserve requirements as a macroeconomic stabilization tool.

A main problem of empirical work on macroprudential policy is how to identify its impact on macroeconomic and financial variables, i.e. how to distinguish correlation and causation. The literature has broadly followed six approaches in dealing with this issue: event studies; assessments of authorities or outside observers; reduced-form regression analysis, typically conducted using cross-country panel regressions; macro stress tests; counterfactual analysis; analysis based on micro data.

One very common approach relies on event studies, typically carried out at central banks (e.g. CGFS, 2010), the IMF (e.g. Hilbers et al, 2005, 2007; Crowe et al, 2011a, b) or the BIS (e.g. Borio and Shim, 2007; Montoro and Moreno, 2011), and covering different cross-section of countries. To give an example, Table 3, taken from Crowe et al (2011a), organizes events studies on the impact of macroprudential measures aimed at dealing with real estate booms. ¹⁴

Using descriptive case studies, Se (2013) suggests that in the aftermath of the Asian crisis, several Asian emerging countries have successfully applied the loan-to-value tool to slow down increases of house prices. Considering country experiences using a mainly descriptive approach, and based on the literature, Terrier et al. (2011) review the use of policy tools to lean against the wind in Latin America, considering microprudential tools, which could be used for macroprudential purposes if calibrated appropriately over the cycle, and macroprudential tools.

¹⁴ Borio et al (2001) provide a discussion of the interaction between practices concerning the valuation of collateral and loan-to-value ratios and an overview of the regulators' experience with LTV ratios.

¹³ CGFS (2012a) discusses the literature on the empirical evidence for the strength of their proposed transmission channels of macroprudential tools, namely for the impact of stress tests, capital or provisioning requirements, sectoral capital requirements, liquidity requirements and asset-side macroprudential instruments on the credit cycle, on resilience and on output.

Table 3 Stylised facts of effect of macroprudential policy responses to real estate booms

Measure	To address	Used in countries	Impact		
Higher/differentiated capital requirements or risk weights by loan type	Rapid credit growth and/or real estate boom	Bulgaria, Croatia, India, Poland, Norway	Not always effective, some side-effects of shifting the risk elsewhere in the system		
Tighter/differentiated loan classification and provisioning requirements	Rapid credit growth and/or real estate boom	Bulgaria, Croatia, Greece, Israel, Ukraine	Limited effect		
Dynamic provisioning	Resilience to cyclical downturn/bust	China, Colombia, India, Spain, Uruguay	So far so good		
Tightening eligibility requirements, e.g. limits on loan-to-value ratios	Real estate boom	China, Hong Kong SAR, Korea, Malaysia, Singapore, Sweden	Short-lived effect on prices and mortgage activity		
Source: Crowe et al. (2011a), Table 2.					

A second approach relies on compiling the assessments of authorities and outside observers on the effectiveness of macroprudential tools (Borio and Shim, 2007). Authorities' assessments provide direct evidence on specific tools. At the same time, they suffer from two main drawbacks. First, they may be subjective assessments of authorities, rather than being based on independent empirical evaluations. Moreover, for individual country experiences there is a problem of the counterfactual – it is not clear what would have happened in the absence of macroprudential measures. In fact, Borio and Shim (2007) note that judging the overall effectiveness of macroprudential tools is hard, and that a more definite answer would need to be based on a more systematic analysis of the different episodes.

A third approach consists of studies that use reduced-form regression analysis. Compared to event studies, this line of research allows more formal tests of the impact of macroprudential tools. However, it suffers from two main drawbacks. First, reduced-form models do not capture well the interaction between macroeconomic, financial and policy variables. Furthermore, since in most cases macroprudential tools have been introduced only in recent years in policymakers' toolkit, there is not much data to assess their effectiveness and transmission channels empirically. Moreover, macroprudential tools have mostly been used in conjunction with other policies, such as monetary policy, so that it is difficult to isolate their effects.

Using cross-country data over the past three decades in panel regressions, Kuttner and Shim (2012) analyse the degree to which macroprudential policy tools have been effective in mitigating housing price and credit cycles, after controlling for country-specific structural factors and macroeconomic variables. They find that decreases in loan-to-value ratios are associated with lower house price growth, and that limiting debt-service-to-income ratios and increasing loan loss provisioning requirements attenuate housing credit growth. Using panel data regressions, Vandenbussche et al. (2012) find that changes in the minimum capital adequacy ratio and non-standard liquidity measures, namely marginal reserve requirements on foreign funding and marginal reserve requirements linked to credit growth, had some impact on house price inflation in Central, Eastern, and South-Eastern European countries, whereas some other macroprudential measures had no impact. Using panel regressions for data from a 2010 IMF survey of the cross-country use of macroprudential policies, Ahuja and Nabar (2011) find that LTV caps decelerate property price growth, that both LTV and DTI caps slow property lending growth, and that LTV caps also affect a broader range of financial stability indicators in economies with pegged exchange rates and currency boards. Wong et al. (2011) find based on panel regressions with data from 13 economies that maximum LTV ratios are effective in

reducing systemic risk stemming from the boom-and-bust cycle of property markets, with the effect on household leverage being more apparent than the effect on property market activity.

Claessens and Ghosh (2012) use a panel data set of credit-, liquidity and capital-related macro-prudential measures and relate these to the pro-cyclicality of three banking variables, namely leverage growth, asset growth, and noncore-to-core liabilities growth. They use a sample of 4673 banks in 170 countries, of which 30 countries implemented at least one macroprudential policy instrument during the period 2000-2010. They consider caps on loan-to-value, caps on debt-to-income, limits on credit growth, reserve requirements, dynamic provisioning, and limits on foreign lending. They find that caps on debt-to-income ratios are relatively effective in emerging economies for a broad range of build-up in financial sector vulnerabilities.

Bruno and Shin (2013) use panel data analysis to study the impact on capital flows of macroprudential policies introduced by Korea in 2010 in comparison with 48 other countries. They find that the sensitivity of capital flows to Korea to global financial conditions decreased following the introduction of macroprudential policies, compared with other countries.

Also using cross-country empirical analysis in panel data regressions, Lim et al. (2011)'s results suggest that macroprudential policy may dampen procyclicality, mainly through credit-related measures (e.g. LTV caps) or liquidity-related measures (e.g. reserve requirements). Also using cross-country data in panel regressions, and consistent with Lim et al. (2011), Dell'Ariccia et al. (2012) find that macroprudential policy tools can reduce the incidence of credit booms and decrease the likelihood that booms end up badly. Using panel fixed-effects models, Wang and Sun (2013) analyse annual data for 171 banks in China from 2010-11, and conclude that some macroprudential policy tools, such as the reserve requirement ratio and housing-related policies, are useful, but cannot guarantee protection against systemic risk; they also suggest that better-targeted macroprudential policies have greater potential to contain systemic risk.

Using cross-country data for Latin American economies in panel regressions, Tovar et al. (2012) tentatively find that reserve requirements had a moderate temporary effect of slowing the pace of credit growth. Using VAR analysis, Federico et al. (2012b) study the macroeconomic effects of changes in legal reserve requirements in Argentina, Brazil, Colombia, and Uruguay. They find that output falls in response to exogenous changes in reserve requirements. Using VAR analysis for Brazil, Glocker and Towbin (2012) find that a discretionary tightening of reserve requirements leads to a decline in domestic credit.

The fourth approach consists of macro stress tests, which can be used to trace the response of the financial system to unusually large exogenous shocks (Sorge, 2004 and Drehmann, 2009). Macro stress tests are by nature forward-looking and highlight the transmission of shocks within the system. They rely explicitly on an underlying view of the forces that can drive financial distress. Similarly to other methodological approaches, however, these models generally fail to capture feedback effects between the financial system and the macroeconomy. They also fail to capture the key aspect of financial distress that small shocks can have very large effects (Borio and Drehmann, 2009).

Recently promising efforts have been undertaken to overcome these shortcomings. Aikman et al (2009) present a Risk Assessment Model for Systemic Institutions (RAMSI) that also includes feedback effects resulting from liquidity risk and embodies an element of procyclicality, which they deem suitable for counterfactual simulations in which regulatory changes – such as varying capital and liquidity buffers – can affect systemic risk. The Bank of Canada's macro stress testing model (Macro-Financial Risk Assessment Framework, MFRAF) estimates systemic risks based on interbank spillover effects at major Canadian banks, linking solvency, market, and funding liquidity risks, and using detailed data on Canadian banks' balance sheets (Gauthier et al. (2010, 2012)). Van den End (2012)'s macro stress-testing model for liquidity risks of banks in the Netherlands incorporates proposed Basel 3 liquidity regulation and second-round feedback effects of shocks, and suggests that Basel 3 liquidity regulation limits liquidity tail risk. Van den End and Kruidhof (2013) use a liquidity

_

¹⁵ BCBS (2013b, 2013c) provide a survey of the macro stress testing literature for liquidity risk; they conclude that liquidity risk modelling still remains in its infancy, especially in macro stress tests.

stress-testing model to simulate the systemic implications of the Basel 3 Liquidity Coverage Ratio (LCR), taking into account the impact of bank reactions on second round feedback effects. They find that a flexible approach to the LCR is a useful macroprudential instrument to mitigate its adverse side-effects during times of stress; but that at extreme stress levels the instrument becomes ineffective and the lender of last resort has to underpin the stability of the system.

A fifth approach conducts counterfactual analysis. Several papers have studied whether macroprudential policy, if it had been applied ahead of the recent financial crisis, could have affected developments positively. Recent unpublished work by the Committee on the Global Financial System (CGFS) of case studies in some major economies finds that, given the knowledge available at the time, macroprudential policy, e.g. via capital or liquidity requirements, could have had positive effects regarding some identified risks in the run-up to the crisis, but not regarding some other risks, e.g. the dependence on cross-currency funding and the growth of the shadow banking system (see FSB/IMF/BIS, 2011a)

Antipa et al. (2010) use a DSGE model estimated for the United States, the euro area and the United Kingdom over the period 1985-2010 to perform counterfactual simulations of macroprudential policies, in the form of an augmented Taylor rule where the short-term interest rate responds to credit growth, and where the authority can increase the cost of credit independently from the short-term interest rate. They find within their model that such macroprudential policy would have been effective in smoothing the last credit cycle and in reducing the depth of the Great Recession. Catte et al. (2010) perform counterfactual simulations within the National Institute of Economic and Social Research's large-scale global macroeconomic model (NiGEM) for the United States for the period 2002-7. They assume that the policymaker has access to an instrument that affects mortgage credit spreads directly, since the model does not allow to specify the nature of macroprudential tools explicitly, with banks not explicitly modelled, for example. They find that use of this instrument would have had a significant effect in reducing real house prices and thereby dampening the housing boom. Barrell et al. (2010) study how macroprudential surveillance could have been better undertaken, using a crisisprediction model for 14 OECD countries estimated over the period 1980-1997. They suggest that countercyclical macroprudential policy would best be calibrated on house prices and current accounts rather than GDP and credit, since they find that credit growth does not have a statistical impact on crisis risk. Izquierdo, Loo-Kung, and Rojas-Suarez (2013) perform simulations for the path of the stock of loan-loss reserves in El Salvador, assuming that the dynamic provisioning rules of Peru or Bolivia had been followed. They suggest that if either of these rules had been followed, Salvadoran banks would have accumulated more provisions during the pre-Lehman period, which could have been used during the financial crisis to support credit.

A sixth empirical approach consists in using micro data, either on individual banks' balance sheet items or on household borrowing. Igan and Kang (2011) for example use data from a survey of mortgages and housing demand by Korean households to examine the impact of LTV and DTI limits on house price dynamics and household leverage in Korea.

There has been particularly limited empirical work on the effects of macroprudential policy tools addressing the cross-sectional dimension, with data limitations being one reason for this. Using data on individual banks' loan books, risk exposures, and on interbank linkages including OTC derivatives for the Canadian banking system, Gauthier et al. (2012) find that macroprudential capital allocation mechanisms reduce default probabilities of individual banks as well as the probability of a systemic crisis by about 25%, suggesting that macroprudential capital buffers can substantially improve financial stability.

BCBS (2013a) provides estimates of the impact of additional capital requirements for globally systemically important banks on growth, based on results from studies by Macroeconomic Assessment Group (2010, 2011). They find that a one percentage point increase in capital for globally systemically important banks would dampen growth on average by an additional 0.07 basis points per year for an eight year implementation period, and by an additional 1.1 basis point per year for a four year implementation period, on average over the transition. Note that the simulated costs refer to the transition to higher capital standards, while the Macroeconomic Assessment Group emphasizes that

benefits in terms of reduced systemic crises are reaped in the steady state. This difference is important. Based on results from BCBS (2010), Macroeconomic Assessment Group estimated that additional capital requirements for globally systemically important banks should provide an annual benefit of about 40–50 basis points of GDP, reflecting the reduced probability of a systemic financial crisis (see BCBS (2013a)).

Miles et al. (2012) provide estimates of socially optimal bank capital, considering overall economic (or social) costs. They calibrate how much increases in bank capital ratios reduce the probability of banking crises and the expected future costs of banking crises. They find that the amount of equity capital which is likely to be desirable for banks is much larger than banks have used in recent years and also higher than targets agreed under the Basel 3 framework, concluding that socially optimal bank capital should be about 20% of risk weighted assets.

The effect of communication about financial stability issues via the publication of Financial Stability Reports and through speeches and interviews over the past 14 years, has been studied in Born et al. (2011). They find that Financial Stability Reports had a significant effect on stock market returns and tended to reduce market volatility. By contrast, speeches and interviews had little effect on market returns and did not reduce volatility during tranquil times, but had a substantial effect during the 2007-10 financial crisis.

5. Issues affecting the effectiveness of macroprudential instruments

For macroprudential policy, one main problem is that regulatory arbitrage, for example through lending via foreign branches or direct cross-border lending, erodes the effectiveness of macroprudential instruments, as discussed for example in Borio and Shim (2007) and Hilbers et al. (2005) based on individual country experiences. For the Eurosystem, this is a particularly relevant issue, which has driven much of the discussions on the European Systemic Risk Board.

Macroprudential policy could also become less effective if risk taking and exposures move outside the regulated banking sector, while remaining systemically important, which calls for coordination with other policy areas. Some historical examples of leakages in regulation are provided in Bank of England (2009), and Hilbers et al. (2005) considered this to be a relevant factor for the recent experience of some CEE economies. Aiyar et al. (2012) find evidence of some leakage associated with foreign bank branches. They find that during 1998-2007, when regulators imposed time-varying bank-specific minimum capital requirements on banks in the United Kingdom, UK-regulated banks reduced lending in response to tighter capital requirements, but non UK-regulated banks (resident foreign branches) increased lending compared with a reference group of regulated banks. Jiménez et al. (2013) find that while countercyclical dynamic provisioning in Spain smoothed cycles in the supply of credit and in bad times upheld firm financing and performance, it did little to stop the credit boom to firms in good times as firms switched to less affected banks.

A lack of coordination of macroprudential policy with monetary policy could also make the former less effective. The literature on the interaction between macroprudential policy and monetary policy is in its infancy, including for example papers of Bean et al. (2010), Angeloni and Faia (2009), Beau et al. (2012), Kannan et al. (2012), and is discussed in Galati and Moessner (2013). Angelini et al. (2012) review the recent literature on the interaction between macroprudential policy and monetary policy. Within a macro stress testing model using data for Dutch banks, van den End (2012) studies the interaction of banks' reactions to Basel 3 liquidity standards with extended refinancing operations and asset purchases by the central bank as part of its unconventional monetary policy, and finds that central banks' asset purchases have more influence on banks relative to refinancing operations due to banks' increased bond holdings. Using panel regressions, Maddaloni and Peydro (2013) find for euro area banks that the impact of low monetary policy rates on the softening of lending standards is reduced by more stringent prudential policy on either bank capital or the loan-to-value ratio for mortgage loans applied in different countries. Federico et al. (2012b) find that reserve requirement policy acted as a substitute for monetary policy rather than a complement. For example, reserve requirements fell in bad times to stimulate output while interest rates increased to prevent a rapid depreciation of the domestic currency. Glocker and Towbin (2012) find that a positive reserve requirement shock leads to an exchange rate depreciation and an improvement in the current account, and to an increase in prices. Their results suggest that reserve requirement policy can complement interest rate policy in pursuing a financial stability objective, but cannot be its substitute with regards to a price stability objective. Tovar et al. (2012)'s results suggest that the effects of changes in policy interest rates and macroprudential policy in the form of reserve requirements have reinforced each other in Latin American economies.

6. Conclusions

The literature on the effectiveness of macroprudential policy tools is still in its infancy and has so far provided only limited guidance for policy decisions. In recent years, however, increasing efforts have been made to fill this gap. Progress has been made in embedding macroprudential policy in theoretical models. There is increasing empirical work on the effect of some macroprudential tools on a range of target variables, such as quantities and prices of credit, asset prices, and on the amplitude of the financial cycle and financial stability. Since empirical evidence on the effectiveness of macroprudential policy tools is still limited, the use of macroprudential policy tools is likely to involve some experimentation, from which authorities setting these instruments can learn over time and which they can use to improve their mix of macroprudential policy tools.

References

Adrian, T. and N. Boyarchenko (2012) Intermediary Leverage Cycles and Financial Stability. Federal Reserve Bank of New York Staff Report No. 56.

Ahuja, A. and M. Nabar (2011) Safeguarding Banks and Containing Property Booms: Cross-Country Evidence on Macroprudential Policies and Lessons from Hong Kong SAR. IMF Working Paper No. 11/284.

Aikman, D., Alessandri, P., Eklund, B., Gai, P., Kapadia, S., Martin, E., Mora, N., Sterne, G. and M. Willison (2009) Funding Liquidity Risk in a Quantitative Model of Systemic Stability. Bank of England Working Paper 372.

Aiyar, S., Calomiris, C. and T. Wieladek (2012) Does macropru leak? Evidence from a UK policy experiment. Bank of England Working Paper No. 445.

Al-Darwish, A., Hafeman, M., Impavido, G., Kemp, M. and P. O'Malley (2011) Possible Unintended Consequences of Basel III and Solvency II. IMF Working Paper WP/11/187.

Angelini, P., Nicoletti-Altimari, S. and I. Visco (2012) Macroprudential, microprudential and monetary policies: conflicts, complementarities and trade-offs. Bank of Italy Occasional Paper No. 140.

Angeloni, I. and E. Faia (2009) A tale of two policies: Prudential regulation and monetary policy with fragile banks. Kiel Working Papers No. 1569, Kiel Institute for the World Economy.

Antipa, P., Mengus, E. and B. Mojon (2010) Would macroprudential policy have prevented the Great Recession?. Manuscript, Banque de France.

Bank of England November (2009) The role of macroprudential policy. A Discussion Paper. Box 7: Leakages in past regulations. November.

Bank of England (2011) Instruments of macroprudential policy. Bank of England Discussion Paper. December.

Barrell, R., Davis, E., Karim, D. and I. Liadze (2010) Calibrating macroprudential policy. NIESR Discussion Paper No. 354.

Basel Committee on Banking Supervision (2010) An assessment of the long-term economic impact of stronger capital and liquidity requirements. Bank for International Settlements, August.

Basel Committee on Banking Supervision (2011) Global systemically important banks: assessment methodology and the additional loss absorbency requirement. Bank for International Settlements, November.

Basel Committee on Banking Supervision (2013a) Global systemically important banks: updated assessment methodology and the higher loss absorbency requirement. Bank for International Settlements, July.

Basel Committee on Banking Supervision (2013b) Liquidity stress testing: a survey of theory, empirics and current industry and supervisory practices. BCBS Working Paper No. 24, October.

Basel Committee on Banking Supervision (2013c) Literature review of factors relating to liquidity stress – extended version. BCBS Working Paper No. 25, October.

Bean, C. (2012) Central banking in boom and slump. Speech at the JSG Wilson Lecture in Economics, University of Hull, Hull, 31 October.

Bean, C., Paustian, M., Penalver, A. and T. Taylor (2010) Monetary policy after the fall. Paper presented at the Federal Reserve Bank of Kansas City Annual Conference, Jackson Hole, Wyoming, 28 August.

Beau, L., Clerc, L. and B. Mojon (2012) Macroprudential policy and the conduct of monetary policy. Banque de France Working Paper No. 390.

Beck, T., Colciago, A. and D. Pfajfar (2014) The Role of Financial Intermediaries in Monetary Policy Transmission. Forthcoming, Journal of Economic Dynamics and Control.

Benes, J., Kumhof, M. and D. Laxton (2014a) Financial Crises in DSGE Models: A Prototype Model. IMF Working Paper WP/14/57, April.

Benes, J., Kumhof, M. and D. Laxton (2014b) Financial Crises in DSGE Models: Selected Applications of MAPMOD. IMF Working Paper WP/14/56, April.

Benigno, G. (2013) Commentary on Macroprudential Policies. International Journal of Central Banking, Vol. 9 (1) March, pp.287-297.

Benigno, G., Chen, H., Otrok, C., Rebucci, A. and E. Young (2013) Financial Crises and Macro-Prudential Policies. Journal of International Economics, Volume 89, Issue 2, March, pp. 453-470.

Bernanke, B., Gertler, M. and Gilchrist, S. (1999) The financial accelerator in a quantitative business cycle framework. In J. Taylor and M. Woodford (eds) *Handbook of Macroeconomics* (Vol. 1, Part C, pp.1341–1393). North Holland: Elsevier.

Bianchi, J. and E. Mendoza (2010) Overborrowing, Financial Crises and 'Macro-prudential' Taxes. NBER Working Paper No. 16091, June 2010, Revised September 2012.

Blanchard, O., Dell'Ariccia, G. and P. Mauro (2010) Rethinking Macroeconomic Policy. IMF Staff Position Note SPN/10/03, 12 February.

Boissay, F., Collard, F. and F. Smets (2013) Booms and systemic banking crises. ECB Working Papers by No. 1514.

Borio, C. (2009) Implementing the macroprudential approach to financial regulation and supervision. Banque de France Financial Stability Review No. 13, September.

Borio, C. (2010) The financial crisis: what implications for new statistics? Keynote speech delivered at the Fifth IFC Conference Initiatives to address data gaps revealed by the financial crisis, Basel, 25–26 August.

Borio, C. and A. Crockett (2000) In search of anchors for financial and monetary stability. Greek Economic Review, 20, 1-14.

Borio, C. and M. Drehmann (2009) Towards an operational framework for financial stability: 'fuzzy' measurement and its consequences. BIS Working Papers, no 284, June.

Borio, C., Furfine, C. and P. Lowe (2001) Procyclicality of the financial system and financial stability: issues and policy options. In Marrying the macro- and micro-prudential dimensions of financial stability. BIS Papers, no 1, March, pp 1–57.

Borio, C. and Shim, I. (2007) What can (macro-)prudential policy do to support monetary policy?. BIS Working Paper No. 242.

Born, B., Ehrmann, M. and M. Fratzscher (2011) Central bank communication on financial stability. ECB Working Paper No. 1332.

Brunnermeier, M., Gorton, G. and A. Krishnamurthy (2014) Liquidity Mismatch. Risk Topography. Chicago: University of Chicago Press, 2014.

Brunnermeier, M., Eisenbach, T. and Y. Sannikov (2013) Macroeconomics with Financial Frictions: A Survey. In Economics And Econometrics, Tenth World Congress Of The Econometric Society. New York: Cambridge University Press.

Brunnermeier, M. and Y. Sannikov (2014) A Macroeconomic Model with a Financial Sector. American Economic Review Vol. 104 (2) pp. 379-421.

Bruno, V. and H. S. Shin (2013) Assessing Macroprudential Policies: Case of Korea. The Scandinavian Journal of Economics, Vol. 116 (1).

Caruana, J. (2010) Macroprudential policy: working towards a new consensus. Remarks at the high-level meeting on The Emerging Framework for Financial Regulation and Monetary Policy jointly organised by the BIS's Financial Stability Institute and the IMF Institute, Washington DC, 23 April.

Catte, P, P Cova, P Pagano and I Visco (2010) The role of macroeconomic policies in the global crisis. Bank of Italy Occasional Papers No. 69, July.

Cecchetti, S., Fender, I. and P. McGuire (2010) Toward a global risk map. BIS Working Paper No. 309, May.

Committee on the Global Financial System (2010) Macroprudential instruments and frameworks: a stocktaking of issues and experiences. CGFS Papers No 38, May.

Committee on the Global Financial System (2012a) Operationalising the selection and application of macroprudential instruments. CGFS Papers No. 48, December.

Committee on the Global Financial System (2012b) Improving the BIS international banking statistics. CGFS Papers No. 47, November.

Cordella, T., Federico, P., Vegh. C. and G. Vuletin (2014) Reserve Requirements in the Brave New Macroprudential World. World Bank Policy Research Working Paper 6793

Crockett, A. (2000) Marrying the micro- and macro-prudential dimensions of financial stability. Remarks before the Eleventh International Conference of Banking Supervisors. Bank for International Settlements. Basel, 20-21 September 2000.

Crowe, C., Dell'Ariccia, G., Igan, D. and P. Rabanal (2011a) How to Deal with Real Estate Booms: Lessons from Country Experiences, IMF Working Paper No. WP/11/91.

Crowe, C., Dell'Ariccia, G., Igan, D. and P. Rabanal (2011b) Policies for Macrofinancial Stability: Options to Deal with Real Estate Booms, IMF Staff Discussion Note No. SDN/11/02.

Cúrdia, V and M Woodford (2010) Credit Spreads and Monetary Policy. Journal of Money, Credit and Banking, vol. 42, pp. 3–35.

de Greef, I.J.M., Hilbers, P.L.C. and L.H. Hoogduin (1996) Moderate Monetarism: A Brief Survey of Dutch Monetary Policy in the Post-war Period. DNB staff reports 28.

De Nicolò, G., G. Favara and L. Ratnovski (2012) Externalities and Macroprudential Policy. IMF staff discussion note 12/05.

Dell'Ariccia, G., Igan, D., Laeven, L. and H. Tong (2012) Policies for macrofinancial stability: How to deal with credit booms. IMF staff discussion note SDN/12/06.

Diamond, D. and P. Dybvig (1983) Bank runs, deposit insurance, and liquidity. Journal of Political Economy 91(3) 401–419.

Diamond, D. and R. Rajan (2001) Banks, short-term debt and financial crises: theory, policy implications and applications. Carnegie-Rochester Conference Series on Public Policy, Elsevier 54(1): 37–71.

Drehmann, M. (2009) Macroeconomic Stress Testing Banks: A Survey of Methodologies. Stress Testing the Banking System: Methodologies and Applications (ed M Quagliariello) Cambridge, Cambridge University Press.

Eichner, M.J., Kohn, D.L. and M.G. Palumbo (2010) Financial Statistics for the United States and the Crisis: What Did They Get Right, What Did They Miss, and How Should They Change?. Finance and Economics Discussion Series 2010-20.

Elliott, D., Feldberg, G. and Lehnert, A. (2013) "The History of Cyclical Macroprudential Policy in the United States. Federal Reserve Board Finance and Economics Discussion Series No. 2013-29.

Englund, P. (1999) The Swedish banking crisis: Roots and consequences. Oxford Review of Economic Policy Vol. 15(3) p. 80-97.

European Systemic Risk Board (2011) Recommendation on the macro-prudential mandate of national authorities. ESRB 2011/3.

European Systemic Risk Board (2014a) The ESRB Handbook on Operationalising Macro-prudential Policy in the Banking Sector.

European Systemic Risk Board (2014b) Flagship Report on Macro-prudential Policy in the Banking Sector.

Federico, P., C. Vegh, and G. Vuletin (2012a) Reserve requirement policy over the business cycle. Manuscript, available at http://www.cass.city.ac.uk/_data/assets/pdf_file/0016/171106/3.-Federico-v2.pdf (accessed 12 October 2013).

Federico, P., Vegh, C. and G. Vuletin (2012b) Effects and role of macroprudential policy: Evidence from reserve requirements based on a narrative approach, manuscript, November, available at http://www.norges-bank.no/pages/90658/Vegh_paper.pdf (accessed 12 October 2013).

Financial Stability Board-bank for International Settlements-International Monetary Fund (2011a) Macroprudential Policy Tools and Frameworks, Progress Report to G20, 27 October.

Financial Stability Board-bank for International Settlements-International Monetary Fund (2011b) Macroprudential tools and frameworks – update to G20 Finance ministers and central bank governors. BIS/FSB/IMF paper, 14 February.

Financial Stability Board-International Monetary Fund (2013) The Financial Crisis and Information Gaps. Fourth Implementation Progress Report of the G20 Data Gaps Initiative. October.

Galati, G. and R. Moessner (2013) Macroprudential policy – a literature review, Journal of Economic Surveys, Vol. 27, No. 5, pp. 846–878.

Gauthier, C., He, Z. and M. Souissi (2010) Understanding systemic risk: the trade-offs between capital, short-term funding and liquid asset holdings. Bank of Canada Working Paper No. 2010-29.

Gauthier, C., Lehar, A. and M. Souissi (2012) Macroprudential capital requirements and systemic risk. Journal of Financial Intermediation, Vol 21, No. 4, pp. 594–618.

Geanakoplos, J. (2011) What's Missing from Macroeconomics: Endogenous Leverage and Default. In Jarocinski, M., F. Smets and C. Thimann (eds.) Approaches to Monetary Policy Revisited – Lesson from the Crisis. Sixth ECB Central Banking Conference 18-19 November 2010, European Central Bank, pp. 220–238.

Gersbach, H. and J.C. Rochet (2012a) Capital Regulation and Credit Fluctuations. CEPR Discussion Paper No. 9077.

Gersbach, H. and J.C. Rochet (2012b) Aggregate Investment Externalities and Macroprudential Regulation. Journal of Money, Credit and Banking, vol. 44, pp. 73-109.

Glocker, C. and P. Towbin (2012) The Macroeconomic effects of reserve requirements. Banque de France Document du Travail No. 374, April.

Goodfriend, M. and McCallum, B. (2007) Banking and interest rates in monetary policy analysis: a quantitative exploration. Journal of Monetary Economics 54, 1480–1507.

Goodhart, C., Kashyap, A., Tsomocos, D. and A. Vardoulakis (2012) Financial Regulation in General Equilibrium. NBER Working Paper No. 17909.

Goodhart, C., Kashyap, A., Tsomocos, D. and A. Vardoulakis (2013) An Integrated Framework for Analyzing Multiple Financial Regulations. International Journal of Central Banking, vol. 9, January.

Grebler, L. (1960) The Selective Credit Controls of 1955. In: NBER, Housing Issues in Economic Stabilization Policy, pp. 37-68.

Group of Thirty (2010) Enhancing Financial Stability and Resilience Macroprudential Policy, Tools, and Systems for the Future. http://www.group30.org/images/PDF/Macroprudential_Report_Final.pdf

Haldane, A. (2011) Risk off. Speech delivered by Andrew G Haldane, Executive Director, Financial Stability and Member of the Financial Policy Committee on 18 August.

Hanson, S., Kashyap, A. and J. Stein (2011) A Macroprudential Approach to Financial Regulation. Journal of Economic Perspectives, 25(1) 3-28.

He, Z. and A. Krishnamurthy (2012) A Model of Capital and Crises. Review of Economic Studies, 79(2)735-777.

He, Z. and A. Krishnamurthy (2013) Intermediary Asset Pricing. American Economic Review. 103(2) 732-770.

He, Z., and Krishnamurthy, A. (2014) A macroeconomic framework for quantifying systemic risk. NBER Working Paper 19885.

Hellwig, M. (1994) Banking and Finance at the End of the Twentieth Century. WWZ Discussion Paper No. 9426, Basel.

Hellwig, M. (1995) Systemic Aspects of Risk Management in Banking and Finance. Swiss Journal of Economics and Statistics 131(4/2) pp. 723-737.

Hilbers, P., Otker-Robe, I., Pazarbasioglu, C. and G. Johnsen (2005) Assessing and managing rapid credit growth and the role of supervisory and prudential policies. IMF Working Papers No. 05/151.

Hilbers, P., Otker-Robe, I. and C. Pazarbasioglu (2007) Analysis of and Policy Responses to Rapid Credit Growth. In *Rapid Credit Growth in Central and Eastern Europe*, eds. Charles Enoch and Inci Otker-Robe, Palgrave MacMillan: New York.

Hodgman, D. (1973) Credit controls in Western Europe: an Evaluative Review. In Credit Control Techniques and Monetary Policy. Conference Series No.11, Federal Reserve Bank of Boston.

Horváth, B. and W. Wagner (2013) The Disturbing Interaction between Countercyclical Capital Requirements and Systemic Risk. Mimeo, Tilburg University.

Houben, A., R. van der Molen and P. Wierts (2012) Making macroprudential policy operational. Revue de Stabilité Financière, 13-25, Banque Centrale du Luxembourg.

Igan, D. and H. Kang (2011) Do Loan-to-Value and Debt-to-Income Limits Work? Evidence from Korea. IMF Working Paper WP/11/297.

International Monetary Fund (2011a) Macroprudential Policy: An Organizing Framework. Prepared by the Monetary and Capital Markets Department.

International Monetary Fund (2011b) Macroprudential Policy: An Organizing Framework. Background Paper. Prepared by the Monetary and Capital Markets Department.

International Monetary Fund (2011c) Regional economic outlook Asia and the Pacific, April.

International Monetary Fund (2013) Key Aspects of Macroprudential Policy.

Izquierdo, A., Loo-Kung, R. and L. Rojas-Suarez (2013) Macroprudential Regulations in Central America. IDB Working Paper series No. IDB-WP-392.

Jaffee, D. and J. Walden (2011) The Impact of Basel III and Solvency 2 on Swedish Banks and Insurers - An Equilibrium Analysis. Financial Market Committee Report no 3, December.

Jeanne, O. and A. Korinek (2010a) Excessive Volatility in Capital Flows: A Pigouvian Taxation Approach. American Economic Review, AEA Papers and Proceedings, pp. 403-407.

Jeanne, O. and A. Korinek (2010b) Managing Credit Booms and Busts: A Pigouvian Taxation Approach. NBER Working Paper 16377.

Jeanne, O. and A. Korinek (2013) Macroprudential Regulation Versus Mopping Up After the Crash. NBER Working Paper No. 18675.

Jiménez, G., Ongena, S., Peydró, J. and J. Saurina (2013) Macroprudential Policy, Countercyclical Bank Capital Buffers and Credit Supply: Evidence from the Spanish Dynamic Provisioning Experiments. European Banking Center Discussion Paper No. 2012-011.

Jonung, L. (1993) The rise and fall of credit controls: the case of Sweden, 1939–89. In Monetary Regimes in Transition, M. Bordo and F. Capie (eds) p. 346-370.

Kannan, P., Rabanal, P. and A. Scott (2012) Monetary and Macroprudential Policy Rules in a Model with House Price Booms. The B.E. Journal of Macroeconomics. Volume 12, Issue 1

Kashyap, A K, D P Tsomocos and A P Vardoulakis (2014) How does macroprudential regulation change bank credit supply?. NBER Working Paper No. 20165.

Kuttner, K. and I. Shim (2012) Taming the Real Estate Beast: The Effects of Monetary and Macroprudential Policies on House Prices and Credit. In A. Heath, F. Packer and C. Windsor (ed.) Property Markets and Financial Stability. Reserve Bank of Australia, pp.231-259.

Lim, C, F Columba, A Costa, P Kongsamut, A Otani, M Saiyid, T Wezel, and X Wu (2011) Macroprudential Policy: What Instruments and How to Use Them? Lessons from Country Experiences. IMF Working Paper 11/238.

Lim, C.H., Krznar, I., Lipinsky, F., Otani, A. and Xiaoyong Wu (2013) The Macroprudential Framework: Policy Responsiveness and Institutional Arrangements. IMF Working Paper WP/13/166.

Lo, A. (2009) The Feasibility of Systemic Risk Measurements. Written testimony for the House Financial Services Committee on Systemic Risk Regulation, October.

Lorenzoni, G. (2008) Inefficient Credit Booms. Review of Economic Studies, July, 75 (3) 809-833.

Macroeconomic Assessment Group (2010) Assessing the macroeconomic impact of the transition to stronger capital and liquidity requirements. Final Report of group established by the Financial Stability Board and the Basel Committee on Banking Supervision, December.

Macroeconomic Assessment Group (2011) Assessment of the macroeconomic impact of higher loss absorbency for global systemically important banks. Report, October.

Maddaloni, A. and J.-L. Peydro (2013) Monetary policy, macroprudential policy and banking stability: evidence from the euro area. International Journal of Central Banking, Vol. 9 No. 1, p. 121-169.

Mahapatra, B. (2012) Underlying concepts and principles of dynamic provisioning. Keynote address at the Conference on Introduction of dynamic provisioning framework for banks in India, organised by the Centre for Advanced Financial Research and Learning (CAFRAL) Mumbai, 21 September.

McCauley, R (2009) Macroprudential policy in emerging markets. Paper presented at the Central Bank of Nigeria's 50th Anniversary International Conference on Central banking, financial system stability and growth, Abuja, 4-9 May.

Miles, D., Yang, J. and G. Marcheggiano (2012) Optimal bank capital. Economic Journal, Vol. 123 (March) p. 1–37.

Moreno, R. (2011) Policymaking from a 'macroprudential' perspective in emerging market economies. BIS Working Paper no. 336.

Montoro, C. and R. Moreno (2011) The use of reserve requirements as a policy instrument in Latin America. BIS Quarterly Review, March, p. 53-65.

Osiński, J., Seal, K. and L. Hoogduin (2013) Macroprudential and Microprudential Policies: Toward Cohabitation. IMF Staff Discussion Paper SDN/13/05.

Perotti, E. and J. Suarez (2011) A Pigovian Approach to Liquidity Regulation. International Journal of Central Banking, vol. 7(4) pp. 3-41, December.

Saurina, J. (2009a) Loan loss provisions in Spain. A working macroprudential tool. Bank of Spain Financial Stability review No. 17, p. 11–26.

Saurina, J (2009b): Dynamic Provisioning. The experience of Spain. Crisis Response. Public Policy for the Private Sector. Note Number 7. July. The World Bank.

Schoenmaker, D. and P. Wierts (2011) Macroprudential Policy: The Need for a Coherent Policy Framework. DSF Policy Paper, No. 13.

Schreft, S (1990) Credit Controls: 1980s. Federal Reserve Bank of Richmond Economic Review, 76(6) pp.25–55.

Se, H. (2013) Loan-to-Value as Macro-Prudential Policy Tool: Experiences and Lessons of Asian Emerging Countries. DSF Policy Paper, No. 33.

Shim, I., Bogdanova, B., Shek, J. and A. Subelyte (2013) Database for policy actions on housing markets. BIS Quarterly Review, September, 83-95.

Shu, C., Su, X., Chiang, J. and H. Miao (2008) China's monetary regime in dealing with large BOP's surpluses: framework, effectiveness and costs. HKMA China Economic Issues, No. 3/08, December.

Sibert, A. (2010) A systemic risk warning system. VoxEU, 16 January.

Simonson, D., and G. Hempel (1993) Banking Lessons from the Past: The 1938 Regulatory Agreement Interpreted. Journal of Financial Services Research, pp. 249-267.

Sorge, M (2004) Stress-testing financial systems: an overview of current methodologies. BIS Working Papers, no 165, December.

Tovar, C., Garcia-Escribano, M. and M. Vera Martin (2012) Credit growth and the effectiveness of reserve requirements and other macroprudential instruments in Latin America. IMF Working Paper WP/12/42.

Tsomocos, D. (2003) Equilibrium analysis, banking, contagion and financial fragility. Bank of England working paper No. 175.

Turner, P. (2009) Currency mismatches and liquidity risk: diagnosis and reform. Paper presented at the EBRD High level workshop on Local currency lending and capital market development in emerging Europe and central Asia. London, 3 December.

Turner, P. (2010) Macroprudential Policies and the Cycle. In 'The Financial Stability Board: An Effective Fourth Pillar of Global Economic Governance?', S Griffith-Jones, E Helleiner and N Woods (eds) Special Report, The Centre for International Governance Innovation, p. 43-48.

Vandenbussche, J., Vogel, U., and Detragiache, E. (2012) Macroprudential Policies and Housing Prices—A New Database and Empirical Evidence for Central, Eastern, and Southeastern Europe. IMF Working Paper No. 12/303.

Van den End, J.W. (2012) Liquidity Stress-Tester: Do Basel III and unconventional monetary policy work?. Applied Financial Economics, Vol 22, No. 15, pp. 1233–57.

Van den End, J.W. (2013) A macroprudential approach to address liquidity risk with the Loan-to-Deposit ratio. DNB Working Paper No. 372, March.

Van den End, J.W. and M. Kruidhof (2013) Modelling the liquidity ratio as macroprudential instrument. Journal of Banking Regulation 14(2) pp.91-106.

Wong, E., Fong, T., Li K., and H. Choi (2011) Loan-to-value ratio as a macro-prudential tool – Hong Kong's experience and cross-country evidence. Hong Kong Monetary Authority Working Paper No. 1.

Previous DNB Working Papers in 2014

- No. 406 **Raymond Chaudron and Jakob de Haan**, Identifying and dating systemic banking crises using incidence and size of bank failures
- No. 407 **Ayako Saiki and Sunghyun Henry Kim**, Business cycle synchronization and vertical trade integration: A case study of the Eurozone and East Asia
- No. 408 Emmanuel de Veirman and Andrew Levin, Cyclical changes in firm volatility
- No. 409 **Carlos Arango, Yassine Bouhdaoui, David Bounie, Martina Eschelbach and Lola Hernández**, Cash management and payment choices: A simulation model with international comparisons
- No. 410 **Dennis Veltrop and Jakob de Haan**, I just cannot get you out of my head: Regulatory capture of financial sector supervisors
- No. 411 Agnieszka Markiewicz and Andreas Pick, Adaptive learning and survey data
- No. 412 **Michael Ehrmann and David-Jan Jansen**, It hurts (stock prices) when your team is about to lose a soccer match
- No. 413 **Richhild Moessner, Jakob de Haan and David-Jan Jansen**, The effect of the zero lower bound, forward guidance and unconventional monetary policy on interest rate sensitivity to economic news in Sweden
- No. 414 Dirk Broeders, An Chen and Birgit Koos, Utility-equivalence of pension security mechanisms
- No. 415 **Irma Hindrayanto, Siem Jan Koopman and Jasper de Winter**, Nowcasting and forecasting economic growth in the euro area using principal components
- No. 416 Richhild Moessner, Effects of ECB balance sheet policy announcements on inflation expectations
- No. 417 Irma Hindrayanto, Jan Jacobs and Denise Osborn, On trend-cycle-seasonal interactions
- No. 418 **Ronald Heijmans, Richard Heuver, Clement Levallois, Iman van Lelyveld,** Dynamic visualization of large transaction networks: the daily Dutch overnight money market
- No. 419 **Ekaterina Neretina, Cenkhan Sahin and Jakob de Haan**, Banking stress test effects on returns and risks
- No. 420 Thorsten Beck, Andrea Colciago and Damjan Pfajfar, The role of financial intermediaries in monetary policy transmission
- No. 421 Carin van der Cruijsen, David-Jan Jansen and Maarten van Rooij, The rose-colored glasses of homeowners
- No. 422 **John Bagnall, David Bounie, Kim Huynh, Anneke Kosse, Tobias Schmidt, Scott Schuh and Helmut Stix**, Consumer cash usage: A cross-country comparison with payment diary survey data
- No. 423 **Ayako Saiki and Jon Frost**, How does unconventional monetary policy affect inequality? Evidence from Japan
- No. 424 **Dirk van der Wal**, The measurement of international pension obligations Have we harmonised enough?
- No. 425 **Ivo Arnold and Saskia van Ewijk**, The impact of sovereign and credit risk on interest rate convergence in the euro area
- No. 426 Niels Vermeer, Maarten van Rooij and Daniel van Vuuren, Social interactions and the retirement age
- No. 427 Richhild Moessner, International spillovers from US forward guidance to equity markets
- No. 428 **Julia Le Blanc, Alessandro Porpiglia, Federica Teppa, Junyi Zhu and Michael Ziegelmeyer,** Household saving behaviour and credit constraints in the Euro area
- No. 429 Lola Hernandez, Nicole Jonker and Anneke Kosse, Cash versus debit card: the role of budget
- No. 430 Natalya Martynova, Lev Ratnovski and Razvan Vlahu, Franchise value and risk-taking in modern banks
- No. 431 **Thorsten Beck, Hans Degryse, Ralph de Haas and Neeltje van Horen**, When arm's length is too far. Relationship banking over the business cycle
- No. 432 Tim de Vries and Jakob de Haan, Credit ratings and bond spreads of the GIIPS
- No. 433 Clemens Bonner, Preferential regulatory treatment and banks' demand for government bonds
- No. 434 Tigran Poghosyan, Charlotte Werger and Jakob de Haan, Size and support ratings of US banks
- No. 435 **Beata Javorcik and Steven Poelhekke**, Former foreign affiliates: Cast out and outperformed?
- No. 436 **Job Boerma**, Openness and the (inverted) aggregate demand logic
- No. 437 **Philip Wilms, Job Swank and Jakob de Haan**, Determinants of the real impact of banking crises: A review and new evidence
- No. 438 **Jacob Bikker and Adelina Popescu**, Efficiency and competition in the Dutch non-life insurance industry: Effects of the 2006 health care reform
- No. 439 Aleš Bulíř, Martin Číhak and David-Jan Jansen, Does the clarity of inflation reports affect volatility in financial markets?

